
BEAST TPC Status
- Humu taken out of FWD tunnel and 

placed in ARICH clean room as spare

- Iiwi put in its place (210Po sources moved 
from Humu)

- HV currents fixed with existing cables 
(matches other TPCs now)

- Communication is stable

- Trigger initially not communicating, now 
fixed

- External trigger still not firing (internal is) 1


